Author's response to reviews

Title: Improvement in health-related quality of life in osteoporosis patients treated with teriparatide

Authors:

Arthur N Lau (arthur.lau@learnlink.mcmaster.ca)
Sammy H Ali (sammy.ali@learnlink.mcmaster.ca)
Anna M Sawka (amsawka@yahoo.com)
Lehana Thabane (thabanel@mcmaster.ca)
Alexandra Papaioannou (PAPAIOANNOU@HHSC.CA)
Amiram Gafni (gafni@mcmaster.ca)
Jonathan D Adachi (jd.adachi@sympatico.ca)

Version: 5 Date: 20 October 2008

Author's response to reviews:

The appropriate formatting changes have been made after reviewing the list of recommendations from the editors.

The abstract has been appropriately changed to meet the appropriate format. A conclusion section has been added. An acknowledgements section has been added and their permission has been obtained. As already stated, this study was not funded in anyway, and financial disclosures have been included. The figure title, and tables are now included in the main text, and the order of tables have been changed to match the order of appearance within the manuscript. The reference section has also been changed to as per the editor's wishes.